Author's response to reviews

**Title:** Investigating the effect of intra-operative infiltration with local anaesthesia on the development of chronic postoperative pain after inguinal hernia repair. A randomized placebo controlled triple blinded and group sequential study design [NCT00484731]

**Authors:**

Philipp Honigmann (philipp.honigmann@ksl.ch)  
Henning Fischer (henning.fischer@ksl.ch)  
Anita Kurmann (anita.kurmann.1@ksl.ch)  
Laurent Audige (laurent.audige@aofoundation.org)  
Guido Schuepfer (guido.schuepfer@ksl.ch)  
Juerg Metzger (juerg.metzger@ksl.ch)

**Version:** 3  
**Date:** 4 November 2007

**Author's response to reviews:**

- figure titles removed from the image files
- figure titles referenced after the table section
- ensured that all figures are mentioned in the text
- citation no. 37 revised
- gap page 14 removed